Author’s response to reviews

Title: A Study Protocol to Investigate the Management of Depression and Challenging Behaviors Associated with Dementia in Aged Care Settings

Authors:

Marita P McCabe Prof (marita.mccabe@deakin.edu.au)
David Mellor Prof (david.mellor@deakin.edu.au)
Tanya E Davison Dr (tanya.davison@monash.edu)
Gery Karantzas Dr (gery.karantzas@deakin.edu.au)
Kathryn von Treuer A/Prof (kathryn.vontreuer@deakin.edu.au)
Daniel W O’Connor Prof (daniel.oconnor@monash.edu)

Version: 2 Date: 26 July 2013

Author’s response to reviews: see over
26 July 2013

Editor
BMC Geriatrics

Dear Editor,

Please find attached the revised manuscript entitled “A Study Protocol to Investigate the Management of Depression and Challenging Behaviors Associated with Dementia in Aged Care Settings” (MS # 1333036470102827). I was not sure where to put them in the manuscript so I put them before the references. I would be most grateful if you would consider this for publication in BMC Geriatrics. This article has not been published previously and is not currently under consideration for publication elsewhere. The authors have no conflict of interest. If you have any queries, please do not hesitate to contact me via e-mail marita.mccabe@deakin.edu.au.

Regards

Marita McCabe PhD FAPS
Alfre Deakin Professor of Psychology